Author's response to reviews

Title: The early postnatal period: exploring women’s views, expectations and experiences of care using focus groups in Victoria, Australia

Authors:

Della A Forster (d.forster@latrobe.edu.au)
Helen L McLachlan (h.mclachlan@latrobe.edu.au)
Jo Rayner (j.rayner@latrobe.edu.au)
Jane Yelland (jane.yelland@mcri.edu.au)
Lisa Gold (lisa.gold@deakin.edu.au)
Sharon Rayner (sharon_rayner@yahoo.com.au)

Version: 6 Date: 11 July 2008

Author's response to reviews:

Please find attached a final revised version with the following formatting changes as requested:

Title page added to front of manuscript

Footnote removed and some of this description incorporated in the main text.

Regards

Della Forster